Author’s response to reviews

Title: Different patterns of myopia prevalence and progression between internal migrant and local resident school children in Shanghai, China: A 2-year cohort study

Authors:

Yingyan Ma (mYy_29@163.com)
Senlin Lin (13681631090@yeah.net)
Jianfeng Zhu (jfzhu1974@hotmail.com)
Xun Xu (drxuxun@sjtu.edu.cn)
Lina Lu (lulina781019@qq.com)
Rong Zhao (zhaorong@smhb.gov.cn)
Qiangqiang Li (263103164@qq.com)
Huijuan Zhao (bsyanfang@sina.com)
Zhiyuan Hou (zyhou@fudan.edu.cn)
Xiangui He (xianhezi@163.com)
Haidong Zou (zouhaidong@hotmail.com)

Version: 2 Date: 07 Feb 2018

Author’s response to reviews:

Dear Editors and Reviewers,

Thank you for your revision of the manuscript. According to your comments, we made several revisions, especially for the editing of grammar, expression and typos.

1. Please have the text edited by a professional language editing service or a native English speaking colleague. There are many issues with grammar, wording, spelling, and/or punctuation that need to be addressed before acceptance for publication.

Response: Thank you for your comments. We have invited a native speaker to revised the grammar, wording, spelling, and punctuation. We can attach a certificate to prove the process if you would like to see.
2. We note that Xun Xu is not listed in the author's contribution section. The individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here:

http://www.biomedcentral.com/submissions/editorial-policies#authorship

Response: Thank you for your comments. Xun Xu participated in the analyses of the data. We have added his contribution in the manuscript.

3. We note that the initials 'JHF' have been included in the Authors' Contribution section. However, these initials do not match any of the authors on the author list. Please clarify whether this was a typing error or if this author needs to be added to your author list.

If you choose to change your author list you will need to fill out a change in authorship form and send it by email to the Editorial office to be approved by the Editor. The form can be found here: https://www.biomedcentral.com/getpublished/editorial-policies#authorship.

Response: Thank you for your comments. JHF is a typo and is not a co-author of the manuscript. We have deleted it.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: Yes, the submitted version is clean.